\begin{table}[t]
	\centering
	\renewcommand{\arraystretch}{1.4}
	\resizebox{\columnwidth}{!}{
		\begin{tabular}{lm{0.2\linewidth}m{0.15\linewidth}m{0.3\linewidth}l}
			\toprule
			Project                                                   & Environment                                             & Stakeholders                                          & Highlighted Features                                                                                                                                                                                                          & GUI                                                                           \\
			\midrule

			CloudSim~\cite{DBLP:journals/spe/CalheirosRBRB11}         & Cloud, Fog~\cite{DBLP:journals/spe/GuptaDGB17}, Edge    & Research                                              & VC\textsuperscript{$\star$}, N, S, E, WF\textsuperscript{$\dagger$}~\cite{DBLP:conf/eScience/ChenD12}, FD\textsuperscript{$\dagger$}, EXP\textsuperscript{$\dagger$}, CM, PI\textsuperscript{$\dagger$}                       & \ding{51}\textsuperscript{$\dagger$}~\cite{DBLP:conf/aina/WickremasingheCB10} \\
			\midrule
			SimGrid~\cite{DBLP:journals/corr/CasanovaGLQS13}          & Grid, P2P, Cloud~\cite{DBLP:journals/fgcs/MichonGGUK17} & Research, Edu~\cite{DBLP:journals/jpdc/CasanovaTKS21} & VC\textsuperscript{$\star$}\textsuperscript{$\dagger$}~\cite{DBLP:journals/tcc/HirofuchiLP18}, N\textsuperscript{$\star$}, S, E\textsuperscript{$\star$}, WF\textsuperscript{$\star$}~\cite{DBLP:data/11/CasanovaSTPJKAOS20f} & \ding{51}\textsuperscript{$\dagger$}~\cite{DBLP:data/11/CasanovaSTPJKAOS20f}  \\
			\midrule
			DGSim~\cite{DBLP:conf/europar/IosupSE08}                  & Grid                                                    & Research                                              & WF, F, EXP                                                                                                                                                                                                                    & \ding{55}                                                                     \\
			\midrule
			GroudSim~\cite{DBLP:conf/europar/OstermannPPF10}          & Grid, Cloud                                             & Research                                              & WF, CM, F                                                                                                                                                                                                                     & \ding{55}                                                                     \\
			\midrule
			iCanCloud~\cite{DBLP:journals/grid/NunezVCCCL12}          & Cloud                                                   & Research                                              & VC, N\textsuperscript{$\star$}, S, CM                                                                                                                                                                                         & \ding{51}\textsuperscript{$\star$}                                            \\
			\midrule
			\textbf{OpenDC}~\cite{DBLP:conf/ccgrid/MastenbroekAJLB21} & Cloud                                                   & Research, Edu.                                        & VC\textsuperscript{$\star$}, N, S, E\textsuperscript{$\star$}, CM, FS\textsuperscript{$\star$}, ML, WF, F\textsuperscript{$\star$}, PI, EXP\textsuperscript{$\star$}                                                          & \ding{51}\textsuperscript{$\star$}                                            \\
			\bottomrule
		\end{tabular}
	}
	\caption{Comparison of selected datacenter simulators. \textbf{Models:} VC = VMs and containers; N = Network, S = Storage, E = Energy, CM = Cost Models, FS = FaaS, ML = Machine Learning, WF = Workflows, FD = Federation; \textbf{Phenomena:} F = Failures, PI = Performance interface; \textbf{Tools:} EXP = Experiment automation; \textbf{Support:} \ding{51} = Yes, \ding{55} = No; $\dagger$ = extension, not integrated; $\star$ = advanced, carefully calibrated feature. Author: Mastenbroek \etal~\cite{DBLP:conf/ccgrid/MastenbroekAJLB21}}
        \label{tab:datacenter_simulator_comparison}
\end{table}

